

# Indian summer monsoon projections under regional and global forcings



**TiPES: Tipping Points in the Earth System** *is is a Research and Innovation action (RIA) funded by the Horizon 2020 Work programme topics "Addressing knowledge gaps in climate science, in support of IPCC reports".* Start date: 1<sup>st</sup> September 2019. End date: 31<sup>st</sup> August 2023.



The TiPES project has received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 820970.

### About this document

**Deliverable:** D3.4 (D13) Work package in charge: WP3 Analysis and modelling of Tipping Elements in future climates Actual delivery date for this deliverable: Project-month 18 (extended to 21) 31st May

**Dissemination level:** The general public (PU)

Lead author(s) University of Reading: Lucy Recchia, Valerio Lucarini

Other contributing author(s)

Reviewer(s)

Visit us on: www.sites.tipes.ku.dk



Access our open access documents in Zenodo: https://zenodo.org/communities/tipes/

Follow us on Twitter: @TiPES\_H2020

**TiPES** Deliverable D3.4

Disclaimer: This material reflects only the author's view and the Commission is not responsible for any use that may be made of the information it contains.

## Index

Summary for publication	5
Work carried out	7
Literature review	8
Results & discussion	18
Main results achieved	40
Progress beyond the state of the art	41
Impact	41
Lessons learned and links built	41
Contribution to the top level objectives of TiPES	42
References (Bibliography)	43
Dissemination and exploitation of TiPES results	50
Dissemination activities	50
Peer reviewed articles	51
Other publications	52
Uptake by targeted audience	52

#### Summary for publication

The aim of this report is to establish the state of the Indian monsoon within future climate scenarios, and to define safe operating spaces to exclude dangerous conditions over the Indian region. This is accomplished through climate modelling studies and comparison of the results to previous studies.

The Indian summer monsoon (hereafter referred to as the Indian monsoon) is a globally significant meteorological event, bringing widespread precipitation annually between May and September. A monsoon consists of a seasonal reversal of wind direction near the surface, changing the large-sale circulation. The Indian monsoon system is particularly complex with respect to other global monsoons, due to the unique topography of the Indian continent with the Himalayas to the north and the oceans to the east and west. Its dynamical nature makes the response to various forcings difficult to predict. Regional and global scale forcings on a range of timescales can impact the Indian monsoon, including internal variability, oceanic phenomena and land-surface processes. The Indian monsoon is strongly coupled with the East Asian monsoon, so a holistic approach that considers the wider effects from both South and East Asia is needed.

In a future climate, land and sea surface temperatures are expected to rise, enabling greater moisture uptake. However, the global circulation is expected to weaken. The net effect is a more intense monsoon with higher rainfall rates, as the thermodynamic response dominates over the dynamic response. Regionally, there is considerable variation in projected rainfall trends. Over southern India, rainfall is expected to increase, whilst over northwest India, it is expected to decrease. Trends for other regions are highly debated. More frequent and intense rainfall events are predicted for India, in line with global predictions. To reduce uncertainty in future climate projections, focus should be given to correctly simulating the timing of monsoon phases (i.e. onset, active/break spells, withdrawal), as well as accurate representation of ocean processes & phenomena.

The Indian monsoon system has been identified as a possible tipping element, highlighting its importance in future global climate. It is thought that if some critical threshold is exceeded, the current state of the Indian monsoon will become unstable, triggering a potentially irreversible transition to a new state. The intensification or breakdown of the Indian monsoon would have far-reaching consequences to the environment and to the economy and from a humanitarian perspective. Changes to the timing, intensity and distribution of rainfall are some of the most crucial aspects to determine. In relation, it is important to establish shifts in the frequency and severity of monsoon depressions. Extreme rainfall events regularly cause floods, endangering livelihoods.

To assess the future response of the Indian monsoon to various factors of a changing climate, simulations are conducted with an intermediate complexity climate model. The roles of absorbing aerosols and greenhouse gases are considered. An absorbing aerosol forcing, applied in the form of mid-tropospheric heating to regions in Asia, causes surface cooling, weakens circulation and suppresses precipitation. Doubling the carbon dioxide concentration partially offsets the aerosol effect, with the associated warming acting to slow the precipitation decline, but leads to further weakening of the circulation.

Whilst increasing aerosol forcing in the regions of India, Southeast Asia and East China leads to decreasing precipitation locally, the forcing over East China is additionally linked to increasing precipitation over India. The net result is an overall decrease in precipitation over India, but the decrease is less than for the other regions. Each region responds differently, in terms of precipitation, to the imposed aerosol forcing. South India is least affected. North India shows a near-linear decline in precipitation as the forcing increases. East China experiences an abrupt transition around 60W to a low precipitation regime. Southeast Asia shows a steeper decrease in precipitation than North India, but without the abrupt transition of East China. As the aerosol forcing is increased to 60W, the precipitable water remains constant, and thereafter decreases. Given that the precipitable water does not decrease in a similar way to the precipitation, it is not a lack of moisture that causes the precipitation decrease, but a reduction in precipitation efficiency.

In a changing climate, the response of the Indian monsoon is uncertain, as there are many factors to consider. Differences in certain aerosol concentrations, altering the radiative forcing, affects both air and sea temperatures. The large-scale atmospheric circulation and cloud processes react to these changes, impacting the development of the monsoon. The situation is complicated as each factor can affect multiple aspects of the monsoon system, sometimes with competing influences. For example, whilst an increase in the amount of carbon dioxide warms the atmosphere, enabling greater moisture uptake and thus a stronger monsoon, an increase in sulphate aerosols acts to suppress the effects. Additionally, global modes of climate variability like El Niño-Southern Oscillation (ENSO) have a strong impact on the Indian Monsoon. Through further modelling studies, the impact of different regional and global forcing scenarios can be better understood and anticipated, facilitating response strategies.

#### Work carried out

This report represents Deliverable 3.4, "Indian summer monsoon projections under regional and global forcings", as part of Work Package 3 ("Analysis and modelling of Tipping Elements in future climates"), for the Horizon 2020 project "Tipping Points in the Earth System" (TiPES). Referring to the project proposal, this deliverable relates to Task 3.3 "Response of the Indian summer monsoon to anthropogenic forcing", which includes the sub-tasks 3.3.1 "Constraining the future evolution of the Indian summer monsoon" and 3.3.2 "Definition of safe operating spaces for the Indian summer monsoon". Our work incorporates Theme 1 "Tipping element in data and models" and Theme 2 "Climate response theory", and contributes to Objective 3 "Characterise climate response in the presence of Tipping Points" and Objective 4 "Define and identify safe operating spaces".

We begin by reviewing the existing literature regarding the future climate response of the Indian monsoon, with a focus on modelling studies. The Indian monsoon system is one of the key tipping points in the Earth's climate and there remains considerable uncertainty in how the system will respond to future climate forcing scenarios. Consolidating the progress made in terms of understanding the complex dynamical interactions within the system, as well as advances in model development, helps inform the direction of research activities. We begin with the IPCC6 report (Intergovermental Panel on Climate Change (IPCC), 2021), followed by several review papers (Ding et al., 2015; UI Hasson et al., 2016; Li et al., 2016; Kitoh, 2017; Naveendrakumar et al., 2019; Hrudya et al., 2020), before focusing on specific model studies. There are few parametric-style studies regarding the Indian monsoon as a tipping element, but a multitude of articles on the future climate of the Indian monsoon in Coupled Model Intercomparison Project Phase (CMIP) 5 and CMIP6 standard models.

To address Task 3.3, we design and implement several simulations with a general climate model to investigate the response of the Indian monsoon to various anthropogenic forcing scenarios. In particular, we determine the conditions under which the monsoon circulation breaks down and the precipitation is drastically reduced. The impact of absorbing aerosols and greenhouse gases, which have opposing effects on the Indian monsoon, is considered. Specifically, we determine the tipping points at which the monsoon almost disappears, by gradually increasing the absorbing aerosol forcing. Additionally, we quantify the extent to which greenhouse gas forcing moderates the absorbing aerosol effect on the monsoon.

#### Literature review

A brief review of the present-day climate is presented, before focusing on predicted changes to the Indian monsoon system in the future climate. An understanding of the present-day internal dynamics and response to external forcings on a range of spatial and temporal scales is required in order to constrain uncertainty in future projections. Relationships between the Indian monsoon and atmospheric and oceanic phenomena are expected to remain robust in the future climate.

#### Present-day climate

An overview of the mechanisms behind the formation of the present-day Indian monsoon is given, as well as a discussion of external factors that are known to influence the monsoon. Established relationships such as the impact of the El Niño-Southern Oscillation (ENSO) on the variability of the monsoon rainfall are highlighted. Understanding the processes and dynamical interactions affecting the Indian monsoon is important for both the present-day and future climate predictions. The section is concluded with a description of observed trends over the last 50 years, focusing on rainfall.

#### Dynamics of the Indian monsoon

The onset of the Indian monsoon is initiated by a seasonal change in the large-scale circulation, setting up a low-level southwesterly flow that brings an influx of moisture from the Arabian Sea. The atmosphere in the southeast becomes favourable for moist convection, encouraging the development of cumulus and congestus clouds and leading to increased rainfall over the region. The Indian monsoon first onsets in the southern region of Kerala in early June, then propagates to the northwest against the mean mid-level wind field. Approximately six weeks after first onset, the entire Indian peninsula experiences the full monsoon, which is associated with unsettled weather and widespread rainfall. The summer monsoon season lasts about three months. In September, the large-scale circulation reverts back to winter conditions and the monsoon withdraws. Figure 1 shows the relative humidity and large-scale circulation at low (850 hPa) and high (200 hPa) levels. Monsoon conditions are represented in the left panels, whilst the right panels show the atmospheric state in winter.

Early explanations of the initiation of the Indian monsoon are based around the classical thermal theory, first presented by Edmund Halley in 1686, which describes the monsoon as a giant sea breeze (Walker, 1972). As the Asian land mass receives more solar radiation and becomes hotter, a temperature gradient develops between the land and the sea, and a heat low forms over north India. Figure 2 illustrates the temperature differences over the land and ocean during the summer monsoon, compared to winter conditions. A vertical pressure gradient is created, with low pressure at the surface and high pressure aloft, modifying the large-scale circulation. Moist air from the Arabian Sea is drawn in over southern India, with convergence at the surface triggering convective processes and leading to rainfall. More recently, the Indian monsoon has been viewed as a regime shift in conjunction with the seasonally migrating Intertropical Convergence Zone (ITCZ) (Bordoni and Schneider, 2008), which





follows the poleward movement of the moist static energy maximum, from the equator (Schneider et al., 2014). Further work is needed to unite the energy budget and dynamic perspective in order to develop a comprehensive monsoon theory (Geen et al., 2020).

#### Influences on the Indian monsoon

There are many influences, both internal and external, that affect the variability of the Indian monsoon. These influences act on a range of timescales, from hours to decades, which adds complexity to the monsoon system. Figure 3 shows the interactions between various influences on the monsoon climate. From a modelling perspec-



Figure 2: Land surface temperature (warm colours, °C) and sea surface temperature (cool colours, °C), for the Indian summer monsoon (June–September) and winter (December–February). Produced from ERA5 reanalysis data (Copernicus Climate Change Service, 2017), averaged over years 1988–2017.

tive, it is difficult to capture the Indian monsoon's response to both fast and slow-acting processes. Continuing research allows for increased understanding of how the components interact with each other and the monsoon system as a whole, enabling improved representation in weather and climate models.

On short timescales, of hours to days, processes that influence the monsoon variability include convection, presence and uptake of moisture, and synoptic disturbances such as easterly waves or low-pressure systems. Changes to the large-scale circulation and intensity of convection on a local scale can affect the spatial distribution of rainfall. The relationship between precipitation and total column water vapour is well documented (Neelin et al., 2009; Schiro and Neelin, 2019). An increase in moisture uptake, from surface evaporation or via advection over the ocean, can hasten monsoon onset and trigger precipitation. Low-pressure systems and monsoon depressions typically bring intense, localised bursts of rain over a short time, which can have a significant impact, particularly with regards to flood risk. It is important to quantify changes to the frequency of occurrence and intensity of these low-pressure systems in a warming climate.

Changes to oceanic properties such as temperature and salinity, and changes to the land surface such as the amount of snow cover and vegetation, affect the heat and moisture fluxes. These changes can occur over weeks–months. Differences in the heat and moisture fluxes close to the ocean/land surface impacts the monsoon response. Other phenomena, acting over years–decades, can also affect oceanic prop-



Figure 3: Schematic adapted from Lau et al. (2000) showing the various components and their contributions to a monsoon climate and its variability.

erties like sea surface temperature. Regional-scale phenomena include the Madden-Julian Oscillation (MJO) and the Quasi-Biweekly Oscillation (QBO). On a global scale there is ENSO, the Indian Ocean Dipole (IOD) and the Tropical Biennial Oscillation (TBO). El Niño years (positive ENSO phase) are linked with a weaker monsoon and La Niña years (negative ENSO phase) are linked with a stronger monsoon. Similarly, a positive IOD is correlated with an increase in monsoon rainfall, whilst a negative IOD is correlated with a decrease in monsoon rainfall. Teleconnections from other ocean basins, for example, the Pacific Decadal Oscillation and the Atlantic Multi-decadal Oscillation, have been linked to the intensity of the monsoon and the total summer rainfall (e.g. Krishnan and Sugi (2003); Hrudya et al. (2020); Naidu et al. (2020)). Such phenomena account for the majority of the interannual variability of the Indian monsoon.

There is more skill in predicting the interannual variability than the intraseasonal variability; however, the intraseasonal variability has a greater impact on the monsoon rainfall. There is value in understanding the relationships between the Indian monsoon and various external forcing factors, as it is likely that robust relationships are maintained in future climate scenarios.

#### Present-day trends

The location, intensity and timing of rainfall are the most key aspects of the monsoon to forecast accurately. In the future climate, changes to rainfall patterns will have the greatest impact. Figure 4 shows the present-day climatology of daily rainfall for the summer monsoon, compared to the winter. Areas receiving the highest rainfall totals



are along the west coast, the Himalayan foothills and northeast India/Bangladesh.

Figure 4: Total daily precipitation (shading, mm/day), for the Indian summer monsoon (June–September) and winter (December–February). Produced from ERA5 reanalysis data (Copernicus Climate Change Service, 2017), averaged over years 1988–2017.

Globally, precipitation is expected to increase with warming surface temperatures. Extreme precipitation events are likely to be more intense and more frequent (Collins et al., 2013; Intergovermental Panel on Climate Change (IPCC), 2013). Over India, there is a decreasing number of monsoon depressions being observed (Dash et al., 2004; Ajayamohan et al., 2010), but an increasing number of weak low-pressure systems (Pattanaik, 2007; Ajayamohan et al., 2010). A "normal" monsoon year is becoming more rare, whilst weaker or stronger monsoon years are becoming more common. The monsoon onset is typically delayed, with the monsoon season peaking in late August/September, rather than July (Turner and Annamalai, 2012). There has also been a spatial shift in the precipitation peak, from east to west, with increased rainfall over the tropical western Pacific and decreased rainfall over south Asia (Annamalai et al., 2013). Despite most models predicting an increase in the total monsoon rainfall over India, observations show a slight negative trend in rainfall from 1950-present day (Figure 5). The negative trend is explained by a dominance of aerosol forcing over greenhouse gases, leading to a drying effect (Bollasina et al., 2011; Kitoh, 2017). The presence of aerosols has an impact on the radiation budget as it causes, by and large, to anomalous heating in the mid troposphere. This leads to an increase in the static stability, which favours the weakening the monsoon circulation, giving a dynamic change of reduced mean moisture convergence (Li et al., 2015). Further work is required to constrain the uncertainty regarding monsoonal rainfall in future climate scenarios.



Figure 5: Historical (1861–1999) and future (2000-2100) projection of South Asian monsoon rainfall under 'balanced' A1B IPCC emission scenario. Time series of mean summer (June–September) precipitation averaged over land points within 60–90° E, 7–27° N. Only four models from the CMIP3 experiment, shown to have a reasonable simulation of the spatial pattern, seasonal cycle and interannual variability of monsoon rainfall, are depicted; the black curve shows their ensemble mean. Observations from the all-India rainfall (AIR) index based on gauge information are also shown for the 1871–2008 period as a proxy for South Asia rainfall. All curves are first normalized by their mean and standard deviation measured over 1961–1999 and are passed through an 11-year moving window. The faint black curve shows the observations without this smoothing. The inset compares the AIR with area-mean averages over the same domain as above from 1951–2004 India Meteorological Department (IMD) daily gridded data and 1901–2009 monthly gridded Climatic Research Unit (CRU) data. The values listed in the legend are for June–September mean rainfall and interannual standard deviation, in mm. Reproduced with permission from Turner and Annamalai (2012).

#### Future climate predictions

Looking ahead to the future 50–100 years, the summer monsoon rainfall is expected to increase (Douville et al., 2000; Cherchi et al., 2011; Turner and Annamalai, 2012; Kitoh, 2017; Krishnan et al., 2020; Intergovermental Panel on Climate Change (IPCC), 2013, 2021). Despite a weakening of the global circulation, the higher atmospheric moisture content due to warmer surface temperatures dominates, leading to a net increase in precipitation (Intergovermental Panel on Climate Change (IPCC), 2021, 2013; Ueda et al., 2006). The inability of models to capture the observed negative rainfall trend in the period 1950–2000 (Saha et al., 2014) reduces confidence in future climate predictions. On the other hand, the substantial number of different climate models predicting an increase in precipitation adds robustness. Complete failure of the Indian monsoon is unlikely (Intergovermental Panel on Climate Change (IPCC), 2013), but lower rainfall years are projected to become more frequent (Schewe and Levermann, 2012; Krishnan et al., 2020). This is in agreement with Turner and Annamalai (2012), in respect to the expectation of monsoon years becoming more severe and less "normal". There remains considerable uncertainty regarding how much the precipitation will increase by, changes to the timing of monsoon phases and effects on a regional scale.

#### Regional trends

There are significant regional differences in projected rainfall in South Asia, which can be lost when considering the net change for all of the sub-continent (Hrudya et al., 2020). Over the southern peninsula, there is general agreement on an increasing rainfall trend (Turner and Annamalai, 2012; Das et al., 2014). There is considerable disagreement over central India and far northeast India. Turner and Annamalai (2012) suggests little change in precipitation over central India; Lau and Kim (2010); Das et al. (2014) suggest a decreasing trend, and Roxy et al. (2017) suggests an increasing trend. Although Das et al. (2014) note a decreasing rainfall trend in the northeast, future projections of tropical rainfall generally follow the pattern of wet-gets-wetter. This means that areas such as Assam, Meghalaya and the Himalayan foothills, which already receive significant amounts of rainfall during the monsoon season, are expected to become wetter Lau and Kim (2010). Similarly, it is anticipated that arid/desert regions, for example Rajasthan, get less rainfall (Turner and Annamalai, 2012; Das et al., 2014). Along the Western Ghats mountain range, decreasing precipitation trends are reported (Das et al., 2014; Varghese et al., 2020). A more detailed study specifically focusing on the Western Ghats region found that whilst rainfall is expected to decrease around the southern end of the Western Ghats range, it increases slightly in the north end of the Western Ghats range. It is important to note that the lack of a comprehensive definition for regions of India makes it difficult to compare regional rainfall trends.

#### Extreme weather events

Over India, there is expected to be a decrease in moderate rainfall events, but an increase in extreme rainfall events (Turner and Annamalai, 2012), in agreement with

the global-scale prediction of more intense and frequent precipitation events occurring (Collins et al., 2013). As noted previously, there is a trend towards fewer monsoon depressions, but instead, a higher number of weaker, low-pressure systems. Supporting this observation, Sandeep et al. (2018) notes a weakening and a poleward shift of low pressure systems originating in the Bay of Bengal. Correspondingly, the frequency of extreme precipitation events over northern India, linked with the low pressure systems, is expected to increase. A rise in severe rain events is also predicted over central India (Srivastava et al., 2016; Roxy et al., 2017).

#### Response to external forcings

The response of the Indian monsoon to anthropogenic forcing is expected to change in the future climate. Historically, aerosol forcing has dominated, explaining the negative trend in precipitation (Dong et al., 2019). Going forward, greenhouse gas forcing is expected to dominate, with the associated thermodynamic effects leading to a more intense monsoon and higher rainfall totals (Ueda et al., 2006; Li et al., 2015; Kitoh, 2017). The cooling effect from sulphate aerosols (Sherman et al., 2021) is generally outweighed by the warming effect from increasing amounts of absorbing aerosols, particularly black carbon. Zhao et al. (2019) additionally links the reduced role of aerosol forcing in a future climate to an increase in extreme precipitation events. Future changes to the distribution of aerosols over the Tibetan Plateau, which plays a key role in monsoon dynamics, are likely to have a significant effect on the monsoonal rainfall, but the response is debated. Further observational and modelling studies are required to understand the relationship between aerosols and the Indian monsoon, as well as to quantify the effects of aerosol-cloud-precipitation interactions (Sanap and Pandithurai, 2015).

The inverse relationship between the Indian monsoon and ENSO has already been noted in a previous section. Future climate simulations suggest more frequent El Niño events, which could trigger more drought or break phases during the monsoon season (Azad and Rajeevan, 2016). Relationships between the Indian monsoon and other oscillations and/or oceanic phenomena are expected to persist in a future climate scenario, but changes to the occurrence and intensity of such phenomena are uncertain. For example, there is little confidence in the future projection of the MJO, and thus the subseasonal monsoon variability, which is linked to the MJO (Intergovermental Panel on Climate Change (IPCC), 2013).

#### Modelling: CMIP6 vs CMIP5

In terms of modelling the Indian monsoon, there has been improvements since the Coupled Model Intercomparison Project Phase 5 (CMIP5), associated with the IPCC Fifth Assessment Report (2013). However, certain biases remain in the current phase, CMIP6, regarding accurate representation of the spatial pattern and intensity of rainfall.

#### Performance of CMIP5

A common issue across CMIP5 models is simulating the timing of the monsoon onset. Typically, onset is late in the simulations compared with observations (UI Hasson et al., 2016). Typically, the monsoon withdrawal is better captured than the onset (Sperber et al., 2013). Many models feature a dry bias over (northern) India, which is linked with a weaker monsoon circulation and an underestimation of the rainfall in the region (Goswami and Goswami, 2017; Wang et al., 2018). The persistence of this dry bias across different models implies a lack of understanding of the physical processes and/or limitations of parameterisation schemes. In contrast, rainfall is often overestimated over the Indian Ocean and East Asia/China. The proportion of large-scale to convective rainfall is often incorrect, with models favouring convective rainfall, whereas it is the large-scale component that dominates in observations (Jain et al., 2019). The air temperature is better simulated than the precipitation (Jain et al., 2019). Sperber et al. (2013) suggests that improving the representation of the 850 hPa winds will improve the distribution of precipitation, and vice versa. Developments in computing may help overcome the challenge of capturing spatial patterns of rainfall, enabling higher resolution model runs which have been shown to reduce the dry bias over India and South Asia (Turner and Annamalai, 2012; Goswami and Goswami, 2017; Chen et al., 2020; Varghese et al., 2020).

Perhaps the most important factor determining accurate simulation of the Indian monsoon is the ability to represent ocean processes and teleconnections (Wang et al., 2018). Annamalai et al. (2017) show that accurate representation of coupled equatorial Indian Ocean processes is key to simulating the monsoon. Wang et al. (2014) highlight the importance of sea surface temperatures in the Atlantic and Pacific Oceans, linking biases in the oceans with biases in monsoonal rainfall. Similarly, the ability of models to reproduce the phases of ENSO is relevant. Unfortunately, CMIP5 models do a poor job of simulating the location and intensity associated with ENSO heating (Sperber et al., 2013). Generally, the multi-model mean of CMIP5 models outperforms any individual model. It is worth noting that CMIP5 models outperform CMIP3 models in all diagnostics (Sperber et al., 2013), showing that our ability to simulate the Indian monsoon is improving.

#### Performance of CMIP6

The Coupled Model Intercomparison Project Phase 6 (CMIP6) predicts a higher rise in precipitation than CMIP5, with CMIP6 models showing a greater sensitivity to greenhouse gas concentrations (Almazroui et al., 2020). There is some agreement between models in CMIP6 regarding changes to regional rainfall totals for the mid–far future, but little agreement for the near-future (Almazroui et al., 2020). Typical biases from CMIP5, such as a dry bias over South Asia and moist biases over the Indian Ocean & East Asia, remain in CMIP6, although the magnitude of these biases has been reduced (McKenna et al., 2020; Chen et al., 2020). Generally, CMIP6 captures the spatial pattern of rainfall better than CMIP5, particularly over northern and central India, likely due the higher horizontal and vertical resolutions and improved parameterisations of convective and microphysical processes (Gusain et al., 2020). Additionally, orographic

precipitation over the Western Ghats and Himalayan foothills is better represented (Gusain et al., 2020). Aspects to be improved include the timing of monsoon onset, number of active/break spells (Gusain et al., 2020), and effects from oceanic phenomena including ENSO, IOD, MJO (see McKenna et al. (2020)). Further work is needed to represent the intensity of different phases of these oscillations and the changes to sea surface temperatures, which impact on the Indian monsoon.

#### **Results & discussion**

The main goal is to understand the impact of aerosol forcing on the regions of India, Southeast Asia and East China, in terms of significant reducing precipitation and weakening the large-scale circulation. We aim to find the level of forcing required to cause the monsoon system to collapse. Additionally, we wish to explore the interaction of regional aerosol forcing with globally increased carbon dioxide concentration.

Here, we describe the experiment design and subsequent simulation results for investigating the response of the Indian monsoon to regional forcing scenarios. Firstly, the model set up and performance is presented, then we show the results from applying an absorbing aerosol-style forcing over Asia. Next, we consider the effect on the monsoons of doubling carbon dioxide levels in combination with absorbing aerosol-style forcing. Finally, the linearity of the system is evaluated.

#### **Model description**

The Planet Simulator (PlaSim) model von Hardenberg (2020); Lunkeit et al. (2011); Fraedrich et al. (2005) is an intermediate complexity climate model, with a dynamical core, parameterised physical processes and a large-scale geostrophic ocean component. It includes parameterisation schemes for land-surface interactions, radiation and convection. There are 10 vertical levels. We use the T42 horizontal resolution, which is approximately 2.8°. For climate science, where the focus is on large-scale features over long timescales, the PlaSim model is an invaluable tool for investigating various future scenarios, at a low computational cost.

There is a precedent for using both the PlaSim model and other models of a similar complexity for climate simulations regarding the Asian monsoons (Wang et al., 2016; Thomson et al., 2021; Herbert et al., 2022). In terms of lower complexity models, Zickfeld et al. (2005) use a box model of the tropical atmosphere to investigate the stability of the Indian monsoon to changes in planetary albedo and carbon dioxide concentration. For higher complexity models, there is an abundance of literature where CMIP5 and CMIP6 standard models are used to explore the response of the Indian monsoon to future climate scenarios (Menon et al., 2013; Li et al., 2015; Kitoh, 2017; Swapna et al., 2018; Varghese et al., 2020; Krishnan et al., 2020; Almazroui et al., 2020; Chen et al., 2020; Moon and Ha, 2020; Wang et al., 2020, 2021; Swaminathan et al., 2022; Intergovermental Panel on Climate Change (IPCC), 2013, 2021). We use a combination of our own simulations with the PlaSim model and results from existing literature that use a hierarchy of models to quantify the Indian monsoon response to a range of future climate scenarios.



Figure 6: Precipitation (left column) and precipitation anomaly (right column) compared to ERA5 reanalysis data for June-July-August (top row) and December-January-February (bottom row).

#### Model validation

A brief evaluation of the PlaSim model is conducted, in order to show that the PlaSim model is capable of reproduces the seasonality of the Indian monsoon to a sufficient degree of accuracy for our purposes. Similar versions of the PlaSim model has shown to perform well in climate simulations (Holden et al., 2016; Platov et al., 2017). Figures 6–8 show the performance of a control simulation with the PlaSim model, compared to ERA5 reanalysis data (Copernicus Climate Change Service, 2017). The PlaSim simulation has been seasonally averaged over a 50 year period, and the ERA5 data has been seasonally averaged over the period 1988–2017. For the following figures, the left columns show the 50-year average of the PlaSim model simulation, and the right columns show the difference between the PlaSim model simulations and the ERA5 data (PlaSim - ERA5).



Figure 7: Surface temperature (left column) and surface temperature anomaly (right column) compared to ERA5 reanalysis data for June-July-August (top row) and December-January-February (bottom row).

There is a clear difference between summer and winter in the PlaSim model simulations, with summer bringing increased precipitation over India and showing the formation of strong low-level southwesterly winds over the Arabian Sea. The PlaSim model underestimates summer precipitation over North India and around the eastern coast of Bay of Bengal, but slightly overestimates precipitation over the Indian Ocean. There is good agreement between the PlaSim model and ERA5 reanalysis data for surface temperature in the region of interest. The low-level southwesterly monsoon flow is stronger in the PlaSim model than in ERA5 data, indicating some inconsistencies in the large-scale circulation. For the purposes of a parametric climate study, it is concluded that the PlaSim model is a sufficiently accurate tool.



Figure 8: Wind speed & direction (left column) and wind speed & direction anomaly (right column) at the 850 hPa level compared to ERA5 reanalysis data for June-July-August (top row) and December-January-February (bottom row).

#### **Experiment design**

To analyse the roles of absorbing aerosols and greenhouse gases, which have contrasting effects, on the Indian monsoon, we implement two forcing scenarios with the PlaSim model: *heat only* and *heat with*  $2xCO_2$ . The *heat only* simulation features increasing absorbing aerosol forcing, whilst the *heat with*  $2xCO_2$  simulation features doubled carbon dioxide levels to represent higher levels of greenhouse gases, as well as increasing absorbing aerosol forcing. This is a simplified version of IPCC 6 forcing scenario SSP3-7.0 (Intergovermental Panel on Climate Change (IPCC), 2021).

The PlaSim model has no explicit treatment of aerosol interactions, so we use midlevel tropospheric heating as a proxy for absorbing aerosols, similarly to Chakraborty et al. (2004). This heating is applied over three vertical levels, approximately corresponding to 550–750 hPa. The heat forcing is applied simultaneously over three regions: India, Southeast Asia and East China, as per Figure 9. We consider the impact of forcing each of these regions in turn, in Section "Sensitivity to area of applied forcing". To maintain radiative balance, the surface is cooled by the same value as the heat forcing.

For both simulations, the heat forcing, used as a proxy for absorbing aerosols, is gradually increased from 0W to 150W, and then decreased back to 0W. This is done over a simulation length of 900 years, giving the rate of applied heating as  $\pm 0.33$ W/year. We consider 30W to be low forcing, 60W medium forcing and 90W high forcing. These values are within observed ranges of radiative forcings (Kumar and Devara, 2012; Vaishya et al., 2018). Heat forcing much above 100W is considered unrealistic in real-world terms, but we want to ensure that we capture the breakdown of the monsoon system.



Figure 9: Regions showing where mid-tropospheric heating has been applied. Shading indicates terrain height (m).

#### Response to absorbing aerosol forcing

The added heat forcing, a proxy for absorbing aerosols, causes the surface to cool in the regions where the forcing is applied, whilst also causing a warm anomaly around 700 hPa. The increased stratification of the atmosphere suppresses precipitation and weakens the large-scale circulation. As the heat forcing is increased, the response of the monsoons become more pronounced.

#### Figure format

The figures in this section are presented as 2 x 6 panels, with the top left panel showing the state of the system at approximately 30W of heating. The remaining five panels show the anomaly with respect to the control simulation (*heat only* - control) at forcings of approximately 30, 60, 90, 120 and 150W. There is no significant hysteresis in the simulations and so and an average is taken of the ascending branch with forcing  $0 \rightarrow 150W$  and the descending branch with forcing  $150 \rightarrow 0W$ . With the exception of 150W, each panel in the figures is produced using the average of the two 10-year means centred around the respective forcing values. The panel for 150W represents the single 10-year period centred on 150W. All means refer only to the summer months - June, July, August. Areas of high orography will be masked in grey for certain pressure levels.

#### Precipitation

Increased heat forcing corresponds with a reduction in precipitation, illustrated by Figure 10 (top two rows). The East China and Southeast Asia regions are the most affected. With the exception of the west coast and northeastern states, areas of higher orography, the majority of India does not follow the trend of declining rainfall. Partly, this is due to the low-level wind, which brings a large influx of moisture from over the Arabian Sea. In contrast, eastern Siberia experiences a reduction in precipitation, despite being outside the heat-forced area. The relationship between aerosol-like forcing and precipitation is complex, with significant regional effects.

#### Precipitable water

Figure 10 (bottom two rows), showing the change in precipitable water with heat forcing, highlights a moist anomaly over the Middle East, and to a lesser extent, the Indian Ocean. Advection of moisture from East to West helps explain why precipitation over India is not reduced as much as over Southeast Asia and East China. It is notable that the precipitable water does not decrease at as fast a rate as the rainfall decreases. Significant reductions in the precipitable water only occur at high rates of forcing, over 90W. Therefore, the decline in rainfall is primarily attributed to a reduction in precipitation efficiency, rather than a scarcity of moisture availability.



Figure 10: Heat only simulation. Contours showing mean decadal June-July-August precipitation (first column, top row) & precipitable water (first column, third row), and mean decadal June-July-August precipitation anomaly (top two rows) & precipitable water anomaly (bottom two rows) compared to the control run, for a range of heat forcing values.



Figure 11: *Heat only* simulation. Contours showing mean decadal June-July-August evaporation (first column, top row) & total mean cloud cover (first column, third row), and mean decadal June-July-August evaporation anomaly (top two rows) & total mean cloud cover anomaly (bottom two rows) compared to the control run, for a range of heat forcing values.



Figure 12: *Heat only* simulation. Contours showing mean decadal June-July-August temperature and mean decadal June-July-August temperature anomaly compared to the control run, for a range of heat forcing values. The top two rows are at the surface and the bottom two rows at 700 hPa. Areas of high orography are masked in grey.



Figure 13: *Heat only* simulation. Contours showing mean decadal June-July-August wind speed & direction (shading & vectors) and mean decadal June-July-August wind speed & direction anomaly (shading & vectors) compared to the control run, for a range of heat forcing values. The top two rows are at 850 hPa and the bottom two rows at 200 hPa. Areas of high orography are masked in grey.

#### Evaporation

The evaporation rate (Figure 11, bottom two rows) decreases with increasing heat forcing, following a similar spatial pattern to the precipitation. However, the decline in evaporation is lesser than the decline in precipitation, meaning that the reduction in rainfall is not solely due to a lack of moisture in the atmosphere. In addition, although the evaporation rate reduces over India, this does not correspond with a reduction in precipitation. Thus, there is a mechanism of moisture being advected to India, likely from the surrounding oceans.

#### Clouds & radiation

In Figure 11, it can be seen that cloud cover over India increases as the forcing increases, further supporting the supposition that the precipitation decline is due to a loss in efficiency, and not a lack of moisture. Generally, cloud cover decreases around coastlines and over ocean, but increases slightly inland to the north and west of India, spatially coherent with the precipitable water. Outgoing longwave radiation (not shown) increases over the Arabian Sea, Southeast Asia and East China, as forcing increases, consistent with the reduction the cloud cover.

#### Temperature

Considering Figure 12 (top two rows), a surface cooling of several degrees is immediately evident over India for approximately 30W of heating. By around 90W of heat forcing, some areas of India and South East Asia have cooled by 10°C. Above 90W, the surface temperature drops more rapidly and becomes unrealistically cool at -15°C in parts of East China when the forcing is close to 150W. At the 700 hPa level (Figure 12, bottom two rows), a warm anomaly develops over East China, which becomes warmer as the forcing increases. There is also a slight warming at 700 hPa over North India. Despite the same forcing being applied over Southeast Asia, there is no corresponding warming at the 700 hPa level. When the heat forcing is greater than 90W, a cold anomaly forms over the Middle East, creating an East-West temperature dipole. There is some hysteresis evident at 700 hPa, with the entire plot region being warmer at the end of the simulation than at the start. The temperature variation at the surface is greater than at the 700 hPa level. In general, the combination of surface cooling and mid-level warming leads to a strong temperature inversion, increasing the static stability of the atmosphere and suppressing moist convective processes.

#### Circulation

The addition of heat forcing, representing absorbing aerosols, weakens the large-scale circulation. At low levels (Figure 13, top two rows), there is a reduction in strength of the southwesterly wind, which is a key driver of both the Indian and East Asian monsoons. With approximately 60W of heating, the wind speed is reduced by 2-3 ms<sup>-1</sup>, and with 90W heating, there is a 4-5 ms<sup>-1</sup> reduction in wind speed. When the maximum forcing is applied, the speed of the southwesterly monsoon wind becomes close to zero. There is a strengthening of the southwesterly wind in East China, causing dry air to be advected towards East Siberia and corresponding to a reduction of precipitation in the region. The mid-level wind field (not shown) experiences changes of a similar magnitude to the low-level wind field. Several regions, including North India, Bangladesh and East China, have increased wind speeds, which are associated with the formation of atmospheric highs. At high levels (Figure 13, bottom two rows), there is a significant reduction in the speed of the Tropical Easterly Jet, from Southeast Asia to the east coast of Somali. Higher heat forcing corresponds with greater declines in easterly wind speeds. Additionally, there is a slight weakening of the westerly subtropical jet, located north of India, at high forcing rates (>90W).

#### Pressure

Several areas - Northwest India, Bangladesh and East China - show increasing height of the 500 hPa level as the forcing increases (Figure 14). The greatest positive geopotential height anomaly, over East China, indicates a region of persistent high pressure. The assertion of an atmospheric high over East China is supported by the mid-level wind field, warmer 700 hPa temperatures and reduced precipitation.



Figure 14: *Heat only* simulation. Contours showing mean decadal June-July-August 500 hPa geopotential height (top left) and mean decadal June-July-August 500 hPa geopotential height anomaly compared to the control run, for a range of heat forcing values. Areas of high orography are masked in grey.

#### Summary & discussion

Increasing mid-tropospheric heating, a proxy for absorbing aerosols, over southern Asia, cools the surface, suppresses precipitation and weakens the large-scale circulation in the region. Areas of high pressure form, most prominently over East China. Some effects, such as the reduction in precipitation, extend to eastern Siberia. There is a much greater decline in precipitation over Southeast Asia and East China, than over India. The weakening of both the Indian and East Asian monsoons in response to absorbing aerosol forcing has been observed and modelled (e.g. Lau and Kim (2010); Bollasina et al. (2011); Ganguly et al. (2012); Song et al. (2014); Dong et al. (2019)). Our simulation results are in agreement with the results of CMIP5 and CMIP6 standard models (Song et al., 2014), in particular with Ayantika et al. (2021) and their historic simulations with the IITM Earth System Model (version 2).

#### Response to combined absorbing aerosol and greenhouse gas forcing

Generally, enhanced carbon dioxide levels leads to higher surface temperatures, higher humidity levels and weakening of the large-scale circulation. The response of the monsoons to greenhouse gas forcing can be contradictory to the response to aerosol forcing. In our experiments, we find that doubling carbon dioxide levels slightly moderates the effect of the aerosol-style heat forcing. For key variables such as precipitation and surface temperature, the decline is lesser for the combination of greenhouse gas and absorbing aerosol forcing than for the absorbing aerosol forcing alone.

#### Figure format

The figures in this section are presented as 2 x 6 panels, with the top row showing the anomaly with respect to the control simulation and the bottom row the anomaly with respect to the *heat only* simulation. The three columns represent the forcing at approximately 30W, 60W and 90W. Given that there is almost no hysteresis, we take the average of the two halves of the dataset; the ascending branch of forcing  $0\rightarrow150W$  and the descending branch of forcing  $150\rightarrow0W$ . Thus, each panel in the figures is produced using the average of the two 10-year means centred around the forcing values of 30, 60 and 90W. All means refer only to the summer months - June, July, August.

#### Precipitation & evaporation

There is a significant decline in precipitation over Southeast Asia and East China as the heat forcing increases, following the pattern of the *heat only* simulation (Figure 15, top two rows). South India and areas of higher orography also see a reduction in rainfall. In contrast, there is a slight increase in precipitation over North India compared to both the control and the *heat only* simulations. Within the forced regions, evaporation rates decline with heat forcing, although at a lesser rate than for the *heat only* simulation. Over the ocean, evaporation is greater for the *heat with*  $2xCO_2$  run, compared with the other simulations.



Figure 15: *Heat with*  $2xCO_2$  simulation. Contours showing mean decadal June-July-August precipitation anomaly (top two rows) & precipitable water anomaly (bottom two rows), for a range of heat forcing values. The top & third rows are the anomaly compared to the control run (*heat with*  $2xCO_2$  - control), and the second & bottom rows are the anomaly compared to *heat only* run (*heat with*  $2xCO_2$  - *heat only*).



Figure 16: *Heat with*  $2xCO_2$  simulation. Contours showing mean decadal June-July-August surface/700 hPa temperature anomaly (top/bottom rows), for a range of heat forcing values. The top & third rows are the anomaly compared to the control run (*heat with*  $2xCO_2$  - control), and the second & bottom rows are the anomaly compared to *heat only* run (*heat with*  $2xCO_2$  - *heat only*). Areas of high orography are masked in grey.



Figure 17: *Heat with*  $2xCO_2$  simulation. Contours showing mean decadal June-July-August 850/200 hPa wind speed & direction anomaly (top/bottom two rows), for a range of heat forcing values. The top & third rows are the anomaly compared to the control run (*heat with*  $2xCO_2$  - control), and the second & bottom rows are the anomaly compared to *heat only* run (*heat with*  $2xCO_2$  - *heat only*). Areas of high orography are masked in grey.

#### Precipitable water

Doubling the carbon dioxide enables greater moisture uptake, thus it is expected that the precipitable water will be higher, as shown by Figure 15 (bottom two rows). Note that the higher levels of precipitable present in the *heat with*  $2xCO_2$  run do not correspond with a significance difference in rainfall between the two heat-forced runs.

#### Temperature

The areas where the heat forcing is applied experience a cooling of the surface, although the surface temperature is  $10-15^{\circ}$ C warmer in the *heat with*  $2xCO_2$  simulation than in the *heat only* simulation. Elsewhere, for the *heat with*  $2xCO_2$  run, both surface and 700 hPa temperatures are around 5°C higher than either the control or the *heat only* runs (Figure 16).

#### Circulation

The large-scale circulation weakens in a similar way to the *heat only* run. At low levels (Figure 17, top two rows), the southwesterly monsoon wind speed decreases in relation to the *heat only* simulation by a further 2-3 ms<sup>-1</sup>. At mid-levels, the anti-cyclonic wind over the Middle East is stronger in the *heat with*  $2xCO_2$  simulation than the control or the *heat only* simulations. As in the *heat only* run, the 500 hPa geopotential height (not shown) indicates areas of high pressure in North India, Bangladesh and East China. Generally, the 500 hPa geopotential height is greater everywhere in the *heat with*  $2xCO_2$  simulation. Considering Figure 17 (bottom two rows), showing the high level wind field, a further weakening of the Tropical Easterly Jet can be seen compared to the *heat only* run. In contrast, there is a strengthening of the westerly Subtropical Jet (30-45°N). The strengthening of the jet is likely related to the warmer mid-tropospheric temperatures in the *heat with*  $2xCO_2$  simulation, analogous to the mechanism described by Rotstayn et al. (2013) for the subtropical jet in the Southern Hemisphere.

#### Consistency of response

The difference between the *heat with*  $2xCO_2$  and the *heat only* simulation remains similar, regardless of the forcing intensity. Looking at the second and bottom rows in Figures 15 to 17, the panels for 30W, 60W and 90W forcing are comparable, indicating linearity in the system.

#### Summary & discussion

In general, greenhouse gases act to warm the surface temperature, enabling greater moisture uptake of the atmosphere and leading to enhanced rainfall (e.g. Douville et al. (2000); Ueda et al. (2006); Cherchi et al. (2011); Samset et al. (2018)), whilst aerosols are responsible for cooling and drying trends (Monerie et al., 2022). We find that enhanced carbon dioxide levels act to partially mitigate the effect of imposed

aerosol-style forcing. Compared to the *heat only* simulation, the *heat with 2xCO*<sub>2</sub> simulation is warmer at the surface and aloft, has higher levels of precipitable water and precipitation, weaker low-level winds but a stronger subtropical jet. Although doubling the carbon dioxide has a significant impact on the Indian monsoon, here we find that the aerosol forcing dominates. The competition between aerosol and greenhouse gas forcing with respect to the Indian monsoon has been explored in a range of modelling experiments (Samset et al., 2018; Wilcox et al., 2020; Ayantika et al., 2021; Swaminathan et al., 2022), yet the uncertainty in the forcing itself limits the degree to which the response can be constrained. Our results suggest that in the future, the anticipated reduction in aerosol concentration will have a greater impact on monsoonal precipitation in India than the increase in greenhouse gases.

#### Sensitivity to area of applied forcing

The location of the applied heat forcing is varied, to investigate both the local regional response and any remote connections. Referring back to Figure 9, a heat forcing of 60W is applied in turn to each of the black-outlined regions - India, Southeast Asia and East China. Figure 18 shows the precipitation and surface temperature anomalies, with respect to the control run, for the three simulations. The data is seasonally averaged over a 50-year period, and we show the summer months: June, July and August.

In general, East China exhibits the greatest response to local forcing, in agreement with the results presented in Section "Response to absorbing aerosol forcing". When either India or Southeast Asia is heat-forced, the spatial response of the region is mostly in accordance with the area of applied forcing. Forcing applied to East China is responsible for the remote connection in Siberia. Furthermore, forcing East China has nearly as great an effect on India as forcing India itself. In particular, it is the forcing over the East China region that is responsible for the increased moisture and cloud cover over India, which is linked with the lesser decline of precipitation over India, compared to the other regions (Figure 18, top row).

Considering the surface temperature anomaly (Figure 18, second row), forcing East China causes the surface temperature to drop in both East China and India. However, forcing Southeast Asia has the opposite effect on India, with a remote link to India leading to surface warming. The contrasting effects on surface temperature in India from forcing East China and Southeast Asia cancel out when the regions are forced simultaneously.



Figure 18: Applying 60W heat forcing in turn to regions of India, Southeast Asia and East China. 50-year June-July-August mean anomaly of labelled variables, compared to the control run. Areas of high orography are masked in grey.

In terms of the circulation, the majority of the changes are attributed to forcing East China (Figure 18, bottom two rows). These changes include a reduction in the low-level southwesterly monsoon flow, increased advection from East China to Siberia at low-mid levels, and a decrease in speed of the Tropical Easterly Jet at high levels. Forcing Southeast Asia also contributes to slowing down the low-level southwesterly wind, which brings moisture from over the Arabian Sea to India. Despite the reduction of moisture influx to India from the southwest, more moisture is retained over India because of the lower rates of advection toward Southeast Asia. This mechanism is primarily attributed to forcing East China.

These results compliment the findings of Herbert et al. (2022), in that varying the aerosol forcing over East China has a greater effect on the surrounding regions than varying forcing over India. Similarly, Guo et al. (2016) find that the biggest contributors to precipitation changes over India are from remote sources.

#### Quantifying the linearity of the response

Combining the separately forced regions of India, Southeast Asia and East China, (Section ) and comparing to the *heat only* simulation at a 60W snapshot, where all the regions are forced simultaneously, we find some linearity in response. Looking at Figure 19, the left and right columns are qualitatively similar. This is in contrast to Herbert et al. (2022), who found a non-linear response when North India and East China were forced separately, compared to being forced simultaneously. The discrepancy is likely due to the use of different models and forcing strategy.

From the earlier figures and sections, it is difficult to quantify the relationship between key variables and the applied forcing. To investigate further, quantities such as precipitation and precipitable water are averaged over several regions and compared to the varying forcing. The linearity of the relationship between the variable and the forcing is considered for the regions of North India, South India, Southeast Asia and East China (as per the black outlined boxes in Figure 9). India is divided into North and South regions because of the variance in response. Figure 20 illustrates the results. Note that each line represents an average of the ascending (forcing  $0 \rightarrow 150W$ ) and the descending (forcing  $150 \rightarrow 0W$ ) halves of the dataset, due to the lack of hysteresis.

#### Relationship between precipitation and forcing

Looking firstly at the precipitation (Figure 20, top), we can see that the convective component dominates. As the forcing increases, the convective precipitation reduces; however, the large-scale precipitation slightly increases. The *heat with*  $2xCO_2$  show higher amounts of convective precipitation for each of the regions. There is significant variation in convective precipitation response between the regions, with East China showing the most abrupt decline and South India being the least impacted. The response of South India to the applied forcing, which causes a cooling and drying effect, is partially mitigated by advection of moisture from the surrounding oceans. For North India, there is a near-linear decline in convective precipitation as the forcing increases. As noted above, East China experiences an abrupt decline at approximately 60W forcing, after which the convective precipitation drops to almost zero. The behaviour of the



Figure 19: Precipitation (top) and surface temperature (bottom). Left column: 10year June-July-August mean anomaly (*heat only* - control) that approximately corresponds to 60W forcing. Right column: sum of anomalies from regionally forced runs (India/East China/SE Asia - control), taken as 50-year June-July-August average.

convective precipitation in Southeast Asia is somewhere between that of East China and North India: it decreases more sharply than North India, but without the abrupt transition at around 60W.

#### Relationship between precipitable water and forcing

The regional variation of area-averaged precipitable water is less than that of precipitation. Rather, the more striking difference is the higher levels of precipitable water in the *heat with CO*<sub>2</sub> runs, compared to the *heat only* runs. The much greater amount of precipitable water in the *heat with CO*<sub>2</sub> runs doesn't correspond to much greater

precipitation. Another key difference between the behaviour of precipitation and precipitable water is that while the precipitation decreases as the forcing increases from 0 to 60W, the *heat only* precipitable water remains constant and the *heat with CO*<sub>2</sub> actually increases slightly. At 60W, there is a clear transition in the precipitable water for all regions, after which the precipitable water declines with further increases in forcing. The reduction in convective precipitation between 0 and 60W forcing is explained by a decrease in precipitation efficiency, and is not due to scarcity of moisture.



Figure 20: Precipitation (top) and precipitable water (bottom), averaged over the regions indicated (following the marked boxes in Figure 9), for *heat only* and *heat with*  $2xCO_2$  runs, against the heat forcing. Precipitation is separated into convective and large-scale components. Variables taken as a running 20-year June-July-August mean.

#### Main results achieved

We have conducted a parametric study with an intermediate complexity climate model, to assess the roles of absorbing aerosol and greenhouse gas forcing on the Indian monsoon. In addition, we have identified the level of regional forcing at which the monsoon breaks down, in terms of a significant reduction in precipitation.

Absorbing aerosol forcing, which we apply through mid-tropospheric heating in our model, causes surface cooling, mid-level warming, weakening circulation and a reduction in (convective) precipitation. Surprisingly, as the forcing increases, the precipitation declines much faster than the precipitable water, indicating that it is a lack of precipitation efficiency, rather than a lack of moisture. Advection of dry air from East China leads to a reduction in precipitation in eastern Siberia, which is outside of the area being forced. Doubling carbon dioxide concentration partially mitigates the effects of the aerosol forcing, through warmer surface temperatures enabling greater moisture take-up, but further weakens the large-scale circulation. On removal of the heat forcing, we find that the monsoon system recovers fully, indicating that there is no hysteresis in our model simulations.

The strongest regional responses, particularly in regards to the circulation, are attributed to aerosol loading over East China. Although the precipitation decline for each region directly corresponds to applying forcing to that region, there is a remote connection between East China and India. Forcing applied to East China leads to an increase in precipitation over India, which is in contrast to the response when forcing is applied to India. When both regions are forced simultaneously, there is a reduction in precipitation over India, but the reduction is much than for Southeast Asia or East China. Comparing simulations where the regions have been forced separately to the simulation where the regions have been forced simultaneously, the results are qualitatively similar, indicating linearity in the response.

We have characterised regional behavioural regimes in terms of area-averaged precipitation and precipitable water. India is separated into North and South regions, due to the significant variance in their responses. South India is the least affected region, likely due to its peninsula nature. North India shows an approximately linear decrease in precipitation in relation to the heat forcing. For East China, there is an sharp transition at around 60W to a regime where the precipitation is close to zero. The precipitation response of Southeast Asia is somewhere between the other regions, with precipitation declining at a faster rate than North India but without the abrupt transition of East China. In terms of the precipitable water, it remains relatively constant for all regions until 60W; thereafter the precipitable water linearly declines with further increases in forcing.

We note the importance of aerosol loading over East China and the competition between aerosol loading over India, in determining the response of the India monsoon to future climate scenarios. A tipping point at approximately 60W of heat forcing causes a shift to a low precipitation regime for the East China region, whilst Southeast Asia and South India show a more linear decline in precipitation with increasing forcing. There is a compensating effect from East China aerosol forcing on precipitation over India. For area-averaged precipitable water, there is a clear transition at 60W for all regions from near-constant to decreasing levels. To maintain a safe operating space for the India monsoon, it is suggested to keep the absorbing aerosol forcing below 60W, through air quality policies and collaboration between Asian countries.

#### Progress beyond the state of the art

With this report we progress beyond the state of the art by using an intermediate complexity climate model to conduct a parametric study regarding the Indian monsoon response to regional and global forcings. There remains considerable uncertainty in the regional response to aerosol forcing over Asia, despite extensive modelling studies. Few studies use intermediate complexity climate models, although they have been shown to perform comparably to CMIP5 and CMIP6 standard models in long climate simulations. To our knowledge, no similar hysteresis-style experiment has been conducted. Furthermore, finding and modelling the breakdown of the Indian monsoon system under regional forcing, in a climate model, is relatively novel.

#### Impact

The Indian monsoon is a key meteorological event, bringing around 80% of India's rainfall during the summer months. Accurate prediction of the Indian monsoon is of great significance for Indian agriculture, industry and the economy. Delayed onset, periods of drought and extreme rainfall events have an impact on millions of people. It is important to understand how the Indian monsoon system will change in an evolving climate. This report reviews the state of the art research on the possible changes to the Indian monsoon system in a future climate, which helps support major scientific assessments such as the IPCC. Our work complements and builds on the existing literature, adding confidence to the results. It is expected that in the future, as greenhouse gases dominate over aerosol forcing, the Indian monsoon will bring more rainfall, despite a weakening in the large-scale circulation. We identify a tipping point in terms of aerosol forcing, beyond which the Indian monsoon breaks down and precipitation drastically declines. We also note the importance of aerosols over East China in their impact on the Indian monsoon. Both of these points have implications for air quality policies.

Regionally, there remains disagreement in the response to future climate scenarios, with the dynamic and thermodynamic reactions disputed. Further work is necessary to address shortcomings and biases in global climate models, which will increase confidence in future projections of the Indian monsoon. An outcome of this report is that it will help guide future modelling strategies of the project partners. It will also contribute to maintaining strong interdisciplinary collaborations across Europe, sustaining leadership and innovation in European climate science.

#### Lessons learned and links built

This report is the first of its kind in terms of assessing a suitable research direction for studying safe operating spaces of the Indian monsoon. We consider the Indian

monsoon in a broader geographical and meteorological context, and identify the importance of regional aerosol forcing which can distinct non-local effects. The literature review presented will be of use to many in the scientific community, specifically experts in tropical meteorology and climate modelling groups active in CMIP. The report will also form the basis of a new publication, Recchia et al. (2022). Our simulations with an intermediate complexity climate model can be used to inform detailed modelling studies with advanced climate models, such as the UK Earth System Model (UKESM).

During the project, links have been built and strengthened with the University of Hamburg, specifically the Institute of Geography (S. UI Hasson) and Institute of Meteorology (F. Lunkeit), through collaboration regarding modelling studies and technical advice. Within the TiPES community, we have strengthened the collaboration to WP3 (including M. Montoya, Universidad Complutense Madrid), by active discussions of the delivery during the weekly online meetings (Gathertown) and 6-monthly updates.

To highlight the importance and implications of this research, we have presented at the European Geosciences Union General Assembly 2022 (see "Dissemination activities") and we plan to present at the American Geophysical Union Fall Meeting 2022.

### Contribution to the top level objectives of TiPES

## Objective 1-Identify tipping elements (TEs) and their interactions in models and data

The Indian monsoon has been identified as a possible tipping element. This report considers the representation of the Indian monsoon system and its dynamical interactions in state of the art climate models. A tipping point is found at 60W of absorbing aerosol style forcing for the East China region, after which the monsoonal precipitation declines to almost zero. The other regions, India and Southeast Asia, show a more linear decrease in precipitation as forcing increases. We also see a transition at 60W from constant to declining precipitable water, for India, Southeast Asia and East China. On removal of the applied forcing, the system fully recovers.

## Objective 3-Characterise climate response in the presence of Tipping Points (TPs)

The response of the Indian monsoon to future climate forcings has been evaluated. Systematic biases in models which contribute to uncertainty in future predictions, have also been noted. A regime shift in the Indian monsoon from high to low precipitation has been discovered when aerosol forcing is applied to the Asian region. Aerosol loading over East China is found to be important for the response of the Indian monsoon.

#### Objective 4-Define and identify safe operating spaces

Having identified a transition at 60W of absorbing aerosol forcing, it is suggested to keep the equivalent aerosol loading under this value in order to maintain a safe operating space. For India, it is important to consider both local and remote aerosol forcing, which can have contrasting effects on the monsoon. Note that increasing levels of carbon dioxide partially negate the negative precipitation impact of aerosol forcing.

#### **References (Bibliography)**

- Ajayamohan, R. S., Merryfield, W. J., and Kharin, V. V. (2010). Increasing trend of synoptic activity and its relationship with extreme rain events over central India. *Journal of Climate*, 23(4):1004–1013.
- Almazroui, M., Saeed, S., Saeed, F., Islam, M. N., and Ismail, M. (2020). Projections of precipitation and temperature over the South Asian countries in CMIP6. *Earth Systems and Environment*, 4(2):297–320.
- Annamalai, H., Hafner, J., Sooraj, K. P., and Pillai, P. (2013). Global Warming Shifts the Monsoon Circulation, Drying South Asia. *Journal of Climate*, 26(9):2701–2718.
- Annamalai, H., Taguchi, B., McCreary, J. P., Nagura, M., and Miyama, T. (2017). Systematic Errors in South Asian Monsoon Simulation: Importance of Equatorial Indian Ocean Processes. *Journal of Climate*, 30(20):8159–8178.
- Ayantika, D. C., Krishnan, R., Singh, M., Swapna, P., Sandeep, N., Prajeesh, A. G., and Vellore, R. (2021). Understanding the combined effects of global warming and anthropogenic aerosol forcing on the South Asian monsoon. *Climate Dynamics*, 56(5):1643–1662.
- Azad, S. and Rajeevan, M. (2016). Possible shift in the ENSO-Indian monsoon rainfall relationship under future global warming. *Scientific Reports*, 6(1):1–6.
- Bollasina, M. A., Ming, Y., and Ramaswamy, V. (2011). Anthropogenic aerosols and the weakening of the South Asian summer monsoon. *science*, 334(6055):502–505.
- Bordoni, S. and Schneider, T. (2008). Monsoons as eddy-mediated regime transitions of the tropical overturning circulation. *Nature Geoscience*, 1(8):515–519.
- Chakraborty, A., Satheesh, S. K., Nanjundiah, R. S., and Srinivasan, J. (2004). Impact of absorbing aerosols on the simulation of climate over the Indian region in an atmospheric general circulation model. *Annales Geophysicae*, 22(5):1421–1434.
- Chen, Z., Zhou, T., Zhang, L., Chen, X., Zhang, W., and Jiang, J. (2020). Global Land Monsoon Precipitation Changes in CMIP6 Projections. *Geophysical Research Letters*, 47(14):e2019GL086902.
- Cherchi, A., Alessandri, A., Masina, S., and Navarra, A. (2011). Effects of increased co<sub>2</sub> levels on monsoons. *Climate Dynamics*, 37(1-2):83–101.
- Collins, M., AchutaRao, K., Ashok, K., Bhandari, S., Mitra, A. K., Prakash, S., Srivastava, R., and Turner, A. (2013). Observational challenges in evaluating climate models. *Nature Climate Change*, 3(11):940–941.
- Copernicus Climate Change Service (2017). ERA5: Fifth generation of ECMWF atmospheric reanalyses of the global climate. *Copernicus Climate Change Service Climate Data Store (CDS).* Accessed 05/05/2020. Available from: https://cds.climate.copernicus.eu/cdsapp#!/home.

- Das, P. K., Chakraborty, A., and Seshasai, M. V. R. (2014). Spatial analysis of temporal trend of rainfall and rainy days during the Indian Summer Monsoon season using daily gridded ( $0.5^{\circ} \times 0.5^{\circ}$ ) rainfall data for the period of 1971–2005. *Meteorological Applications*, 21(3):481–493.
- Dash, S. K., Kumar, J. R., and Shekhar, M. S. (2004). On the decreasing frequency of monsoon depressions over the Indian region. *Current Science*, 86(10):1404–1411.
- Ding, Y., Liu, Y., Song, Y., and Zhang, J. (2015). From MONEX to the global monsoon: A review of monsoon system research. *Advances in Atmospheric Sciences*, 32(1):10–31.
- Dong, B., Wilcox, L. J., Highwood, E. J., and Sutton, R. T. (2019). Impacts of recent decadal changes in Asian aerosols on the East Asian summer monsoon: roles of aerosol-radiation and aerosol-cloud interactions. *Climate Dynamics*, 53(5):3235– 3256.
- Douville, H., Royer, J. F., Polcher, J., Cox, P., Gedney, N., D. B., S., and P. J., V. (2000). Impact of CO<sub>2</sub> Doubling on the Asian Summer Monsoon: Robust Versus Model-Dependent Responses. *Journal of the Meteorological Society of Japan. Ser. II*, 78(4):421–439.
- Fraedrich, K., Jansen, H., Kirk, E., Luksch, U., and Lunkeit, F. (2005). The Planet Simulator: Towards a user friendly model. *Meteorologische Zeitschrift*, 14(3):299–304.
- Ganguly, D., Rasch, P. J., Wang, H., and Yoon, J.-h. (2012). Fast and slow responses of the South Asian monsoon system to anthropogenic aerosols. *Geophysical Research Letters*, 39(18).
- Geen, R., Bordoni, S., Battisti, D. S., and Hui, K. L. (2020). Monsoons, ITCZs and the Concept of the Global Monsoon. *Earth and Space Science Open Archive*, page 60.
- Goswami, B. B. and Goswami, B. N. (2017). A road map for improving dry-bias in simulating the South Asian monsoon precipitation by climate models. *Climate Dynamics*, 49(5-6):2025–2034.
- Guo, L., Turner, A. G., and Highwood, E. J. (2016). Local and remote impacts of aerosol species on Indian summer monsoon rainfall in a GCM. *Journal of Climate*, 29(19):6937–6955.
- Gusain, A., Ghosh, S., and Karmakar, S. (2020). Added value of CMIP6 over CMIP5 models in simulating Indian summer monsoon rainfall. *Atmospheric Research*, 232:104680.
- Herbert, R., Wilcox, L. J., Joshi, M., Highwood, E., and Frame, D. (2022). Nonlinear response of Asian summer monsoon precipitation to emission reductions in South and East Asia. *Environmental Research Letters*, 17(1):014005.

- Holden, P. B., Edwards, N. R., Fraedrich, K., Kirk, E., Lunkeit, F., and Zhu, X. (2016). PLASIM–GENIE v1. 0: a new intermediate complexity AOGCM. *Geoscientific Model Development*, 9(9):3347–3361.
- Hrudya, P. H., Varikoden, H., and Vishnu, R. (2020). A review on the Indian summer monsoon rainfall, variability and its association with ENSO and IOD. *Meteorology and Atmospheric Physics*, pages 1–14.
- Intergovermental Panel on Climate Change (IPCC) (2013). IPCC Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. [Stocker, T. F., D. Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P. M. Midgley (eds.)]. Cambridge University Press.
- Intergovermental Panel on Climate Change (IPCC) (2021). IPCC, 2021: Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. [Masson-Delmotte, V., P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M. I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J. B. R. Matthews, T. K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.)]. Cambridge University Press.
- Jain, S., Salunke, P., Mishra, S. K., and Sahany, S. (2019). Performance of CMIP5 models in the simulation of Indian summer monsoon. *Theoretical and Applied Climatology*, 137(1):1429–1447.
- Kitoh, A. (2017). The Asian Monsoon and its Future Change in Climate Models: A Review. *Journal of the Meteorological Society of Japan. Ser. II.*
- Krishnan, R., Sanjay, J., Gnanaseelan, C., Mujumdar, M., Kulkarni, A., and Chakraborty, S. (2020). Assessment of Climate Change over the Indian Region: A Report of the Ministry of Earth Sciences (MoES), Government of India. Springer Nature.
- Krishnan, R. and Sugi, M. (2003). Pacific decadal oscillation and variability of the Indian summer monsoon rainfall. *Climate Dynamics*, 21(3-4):233–242.
- Kumar, S. and Devara, P. C. S. (2012). A long-term study of aerosol modulation of atmospheric and surface solar heating over Pune, India. *Tellus B: Chemical and Physical Meteorology*, 64(1):18420.
- Lau, K. M., Kim, K. M., and Yang, S. (2000). Dynamical and boundary forcing characteristics of regional components of the Asian summer monsoon. *Journal of Climate*, 13(14):2461–2482.
- Lau, W. K. M. and Kim, K.-M. (2010). Fingerprinting the impacts of aerosols on longterm trends of the Indian summer monsoon regional rainfall. *Geophysical Research Letters*, 37(16).

- Li, X., Ting, M., Li, C., and Henderson, N. (2015). Mechanisms of Asian Summer Monsoon Changes in Response to Anthropogenic Forcing in CMIP5 Models. *Journal of Climate*, 28(10):4107–4125.
- Li, Z., Lau, W. K.-M., Ramanathan, V., Wu, G., Ding, Y., Manoj, M. G., Liu, J., Qian, Y., Li, J., Zhou, T., Fan, J., Rosenfeld, D., Ming, Y., Wang, Y., Huang, J., Wang, B., Xu, X., Lee, S.-S., Cribb, M., Zhang, F., Yang, X., Zhao, C., Takemura, T., Wang, K., Xia, X., Yin, Y., Zhang, H., Guo, J., Zhai, P. M., Sugimoto, N., Babu, S. S., and Brasseur, G. P. (2016). Aerosol and monsoon climate interactions over Asia. *Reviews of Geophysics*, 54(4):866–929.
- Lunkeit, F., Borth, H., Bottinger, M., Fraedrich, K., Jansen, H., Kirk, E., Kleidon, A., Luksch, U., Paiewonsky, P., Schubert, S., Sielmann, F., and Wan, H. (2011). *Planet Simulator Reference Manual, Version 16.* Meteorological Institute of the University of Hamburg, Hamburg, Germany.
- McKenna, S., Santoso, A., Gupta, A. S., Taschetto, A. S., and Cai, W. (2020). Indian Ocean Dipole in CMIP5 and CMIP6: characteristics, biases, and links to ENSO. *Scientific Reports*, 10(1):1–13.
- Menon, A., Levermann, A., Schewe, J., Lehmann, J., and Frieler, K. (2013). Consistent increase in Indian monsoon rainfall and its variability across CMIP-5 models. *Earth System Dynamics*, 4(2):287–300.
- Monerie, P.-A., Wilcox, L. J., and Turner, A. G. (2022). Effects of anthropogenic aerosol and greenhouse gas emissions on Northern Hemisphere monsoon precipitation: mechanisms and uncertainty. *Journal of Climate*, pages 1–66.
- Moon, S. and Ha, K.-J. (2020). Future changes in monsoon duration and precipitation using CMIP6. *npj Climate and Atmospheric Science*, 3(45):1–7.
- Naidu, P. D., Ganeshram, R., Bollasina, M. A., Panmei, C., Nürnberg, D., and Donges, J. F. (2020). Coherent response of the Indian Monsoon Rainfall to Atlantic Multi-decadal Variability over the last 2000 years. *Scientific Reports*, 10(1):1–11.
- Naveendrakumar, G., Vithanage, M., Kwon, H.-H., Chandrasekara, S. S. K., Iqbal, M. C. M., Pathmarajah, S., Fernando, W. C. D. K., and Obeysekera, J. (2019). South Asian perspective on temperature and rainfall extremes: A review. *Atmospheric Research*, 225:110–120.
- Neelin, J. D., Peters, O., and Hales, K. (2009). The transition to strong convection. *Journal of the Atmospheric Sciences*, 66(8):2367–2384.
- Pattanaik, D. R. (2007). Analysis of rainfall over different homogeneous regions of India in relation to variability in westward movement frequency of monsoon depressions. *Natural Hazards*, 40(3):635–646.
- Platov, G., Krupchatnikov, V., Martynova, Y., Borovko, I., and Golubeva, E. (2017). A new earth's climate system model of intermediate complexity, PlaSim-ICMMG-1.0:

description and performance. In *IOP Conference Series: Earth and Environmental Science*, volume 96, page 012005. IOP Publishing.

- Recchia, L. G., Lucarini, V., Lunkeit, F., and ul Hasson, S. (2022). Effect of varying aerosol forcing on the Indian summer monsoon in an intermediate complexity climate model. *To be submitted*.
- Rotstayn, L. D., Collier, M. A., Jeffrey, S. J., Kidston, J., Syktus, J. I., and Wong, K. K. (2013). Anthropogenic effects on the subtropical jet in the Southern Hemisphere: aerosols versus long-lived greenhouse gases. *Environmental Research Letters*, 8(1):014030.
- Roxy, M. K., Ghosh, S., Pathak, A., Athulya, R., Mujumdar, M., Murtugudde, R., Terray,
  P., and Rajeevan, M. (2017). A threefold rise in widespread extreme rain events over central India. *Nature Communications*, 8(1):1–11.
- Saha, A., Ghosh, S., Sahana, A. S., and Rao, E. P. (2014). Failure of CMIP5 climate models in simulating post-1950 decreasing trend of Indian monsoon. *Geophysical Research Letters*, 41(20):7323–7330.
- Samset, B. H., Sand, M., Smith, C. J., Bauer, S. E., Forster, P. M., Fuglestvedt, J. S., Osprey, S., and Schleussner, C.-F. (2018). Climate Impacts From a Removal of Anthropogenic Aerosol Emissions. *Geophysical Research Letters*, 45(2):1020–1029.
- Sanap, S. D. and Pandithurai, G. (2015). The effect of absorbing aerosols on Indian monsoon circulation and rainfall: A review. *Atmospheric Research*, 164:318–327.
- Sandeep, S., Ajayamohan, R. S., Boos, W. R., Sabin, T. P., and Praveen, V. (2018). Decline and poleward shift in Indian summer monsoon synoptic activity in a warming climate. *Proceedings of the National Academy of Sciences*, 115(11):2681–2686.
- Schewe, J. and Levermann, A. (2012). A statistically predictive model for future monsoon failure in India. *Environmental Research Letters*, 7(4):044023.
- Schiro, K. A. and Neelin, J. D. (2019). Deep Convective Organization, Moisture Vertical Structure, and Convective Transition Using Deep-Inflow Mixing. *Journal of the Atmospheric Sciences*, 76(4):965–987.
- Schneider, T., Bischoff, T., and Haug, G. H. (2014). Migrations and dynamics of the intertropical convergence zone. *Nature*, 513:45–53.
- Sherman, P., Gao, M., Song, S., Archibald, A. T., Abraham, N. L., Lamarque, J.-F., Shindell, D., Faluvegi, G., and McElroy, M. B. (2021). Sensitivity of modeled Indian monsoon to Chinese and Indian aerosol emissions. *Atmospheric Chemistry and Physics*, 21(5):3593–3605.
- Song, F., Zhou, T., and Qian, Y. (2014). Responses of East Asian summer monsoon to natural and anthropogenic forcings in the 17 latest CMIP5 models. *Geophysical Research Letters*, 41(2):596–603.

- Sperber, K. R., Annamalai, H., Kang, I.-S., Kitoh, A., Moise, A., Turner, A., Wang, B., and Zhou, T. (2013). The asian summer monsoon: an intercomparison of CMIP5 vs. CMIP3 simulations of the late 20th century. *Climate Dynamics*, 41(9-10):2711–2744.
- Srivastava, A. K., Singh, G. P., and Singh, O. P. (2016). Variability and trends in extreme rainfall over India. *Mausam*, 67(4):745–766.
- Swaminathan, R., Parker, R. J., Jones, C. G., Allan, R. P., Quaife, T., Kelley, D. I., De Mora, L., and Walton, J. (2022). The Physical Climate at Global Warming Thresholds as Seen in the U.K. Earth System Model. *Journal of Climate*, 35(1):29–48.
- Swapna, P., Krishnan, R., Sandeep, N., Prajeesh, A. G., Ayantika, D. C., Manmeet, S., and Vellore, R. (2018). Long-Term Climate Simulations Using the IITM Earth System Model (IITM-ESMv2) With Focus on the South Asian Monsoon. *Journal of Advances in Modeling Earth Systems*, 10(5):1127–1149.
- Thomson, J. R., Holden, P. B., Anand, P., Edwards, N. R., Porchier, C. A., and Harris, N. B. W. (2021). Tectonic and climatic drivers of Asian monsoon evolution. *Nature communications*, 12(1):1–10.
- Turner, A. G. and Annamalai, H. (2012). Climate change and the South Asian summer monsoon. *Nature Climate Change*, 2(8):587–595.
- Ueda, H., Iwai, A., Kuwako, K., and Hori, M. E. (2006). Impact of anthropogenic forcing on the Asian summer monsoon as simulated by eight GCMs. *Geophysical Research Letters*, 33(6).
- Ul Hasson, S., Pascale, S., Lucarini, V., and Böhner, J. (2016). Seasonal cycle of precipitation over major river basins in South and Southeast Asia: A review of the CMIP5 climate models data for present climate and future climate projections. *Atmospheric Research*, 180:42–63.
- Vaishya, A., Babu, S. N. S., Jayachandran, V., Gogoi, M. M., Lakshmi, N. B., Moorthy, K. K., and Satheesh, S. K. (2018). Large contrast in the vertical distribution of aerosol optical properties and radiative effects across the Indo-Gangetic Plain during the SWAAMI–RAWEX campaign. *Atmospheric Chemistry and Physics*, 18(23):17669–17685.
- Varghese, S. J., Surendran, S., Rajendran, K., and Kitoh, A. (2020). Future projections of Indian Summer Monsoon under multiple RCPs using a high resolution global climate model multiforcing ensemble simulations. *Climate Dynamics*, 54(3):1315– 1328.
- von Hardenberg, J. (2020). General Circulation Models Planet Simulator (PlaSim) and PUMA. *Forked from HartmutBorth/PLASIM.* Accessed 11/05/2022. Available from: https://github.com/jhardenberg/PLASIM.
- Walker, J. M. (1972). The monsoon of Southern Asia: A review. *Weather*, 27(5):178–189.

- Wang, B., Biasutti, M., Byrne, M. P., Castro, C., Chang, C.-P., Cook, K., Fu, R., Grimm,
  A. M., Ha, K.-J., Hendon, H., Kitoh, A., Krishnan, R., Lee, J.-Y., Li, J., Liu, J., Moise,
  A., Pascale, S., Roxy, M. K., Seth, A., Sui, C.-H., Turner, A., Yang, S., Yun, K.-S.,
  Zhang, L., and Zhou, T. (2021). Monsoons Climate Change Assessment. *Bulletin of the American Meteorological Society*, 102(1):E1–E19.
- Wang, B., Jin, C., and Liu, J. (2020). Understanding Future Change of Global Monsoons Projected by CMIP6 Models. *Journal of Climate*, 33(15):6471–6489.
- Wang, C., Zhang, L., Lee, S.-K., Wu, L., and Mechoso, C. R. (2014). A global perspective on CMIP5 climate model biases. *Nature Climate Change*, 4(3):201–205.
- Wang, Z., Li, G., and Yang, S. (2018). Origin of Indian summer monsoon rainfall biases in CMIP5 multimodel ensemble. *Climate Dynamics*, 51(1):755–768.
- Wang, Z., Zhang, H., and Zhang, X. (2016). Projected response of East Asian summer monsoon system to future reductions in emissions of anthropogenic aerosols and their precursors. *Climate Dynamics*, 47(5):1455–1468.
- Wilcox, L. J., Liu, Z., Samset, B. H., Hawkins, E., Lund, M. T., Nordling, K., Undorf, S., Bollasina, M., Ekman, A. M. L., Krishnan, S., Merikanto, J., and Turner, A. G. (2020). Accelerated increases in global and Asian summer monsoon precipitation from future aerosol reductions. *Atmospheric Chemistry and Physics*, 20(20):11955– 11977.
- Zhao, A. D., Stevenson, D. S., and Bollasina, M. A. (2019). The role of anthropogenic aerosols in future precipitation extremes over the Asian Monsoon Region. *Climate Dynamics*, 52(9):6257–6278.
- Zickfeld, K., Knopf, B., Petoukhov, V., and Schellnhuber, H. J. (2005). Is the Indian summer monsoon stable against global change? *Geophysical Research Letters*, 32(15).

## **Dissemination and exploitation of TiPES results**

#### **Dissemination activities**

Type of dissemination activity	Name of the scientist (institution), title of the presenta- tion, event	Place and date of the event	Estim- ated bud- get	Type of audience	Estimated number of persons reached	Link to Zen- odo up- load	
Communication in the host institution & social media engagement	Department of Mathe- matics and Statistics, Centre for the Mathe- matics of Planet Earth, University of Reading	ongoing	N/A	Scientific community (higher education, research)	150	N/A	
EGU presentation	Lucy Recchia (UR), "Ef- fect of varying aerosol forcing on the Indian summer monsoon in an intermediate complex- ity climate model"	online, 24/05/22	£160	Scientific commu- nity (research)	200	https://m 12436.ht	eetingorganizer.c ml
TiPES presentation	Lucy Recchia (UR), " Investigating the response of the In- dian & East Asian monsoons to varying anthropogenic aerosol forcing, in accordance with future climate scenarios"	online, 22/04/22	N/A	TiPES project	50	N/A	
TiPES presentation	Lucy Recchia (UR), WP3 update	online, Jan 22	N/A	TiPES project	30	N/A	
TiPES presentation	Lucy Recchia (UR), WP3 update	online, Jun 21	N/A	TiPES project	30	N/A	
TiPES presentation	Lucy Recchia (UR), WP3 update	online, Sep 21	N/A	TiPES project	30	N/A	

#### Peer reviewed articles

Title	Authors	Publica- tion	DOI	Is TiPES cor- rectly ac- knowl- edged?	How much did you pay for the pub- lica- tion?	Status?	Open ac- cess grante	Comme on the em- bargo im- posed by the pub- lisher	If in Green OA, pro- vide the link where this pub- lica- tion can be found
Controls on propagation of the Indian monsoon onset in an idealised model	L.G. Rec- chia, S.D. Griffiths, D.J. Parker	Quarterly Journal of the Royal Meteo- rological Society	10.1002/ qj.4165	Yes		Published	Yes	None	N/A
Effect of varying aerosol forcing on the Indian summer monsoon in an in- termediate complex- ity climate model	L.G. Rec- chia, V. Lucarini, F. Lunkeit, S. ul Hasson	Earth System Dynamics	N/A	Yes	N/A	In prep.	N/A	N/A	N/A
Seasonal prediction of Indian summer monsoon onset with echo state networks	T. Mitsui, N. Boers	Environ- mental Research Letters	10.1088/ 1748-9326/ ac0acb	Yes		Published	Yes	None	N/A
Holocene climate forc- ings and lacustrine regime shifts in the Indian summer monsoon realm	S. Prasad, N. Marwan, D. Eroglu, B. Goswami, P.K. Mishra, B. Gaye, A. Anoop, N. Basavaiah, M. Stebich, A. Jehangir	Earth Surface Pro- cesses and Land- forms	10.1002/ esp.5004	Yes		Published	Yes	None	N/A

-----

\_

#### **Other publications**

https://www.tipes.dk/improved-prediction-of-indian-summer-monsoon-onset-three-months-in-advance-using-machine-learning/

#### Uptake by targeted audiences

As indicated in the Description of the Action, the audience for this deliverable is:

X	The general public (PU) and is made available to the world via https://cordis.europa.eu/.
	The project partners, including the Commission services (PP).
	A group specified by the consortium, including the Commission services (RE).
	This report is confidential, only for members of the consortium, including the
	Commission services (CO).